Author's response to reviews

Title: Microalbuminuria among Type 1 and Type 2 diabetic patients of African origin in Dar Es Salaam, Tanzania

Authors:

Janet JK Lutale (janet.lutale@rasmus.uib.no)
Hrafnkell B Thordason (hraf@haukeland.no)
Zulfiqarali G Abbas (zabbas@muchs.ac.tz)
Kaare R Vetvik (kare.vetvik@meda.uib.no)

Version: 3 Date: 26 December 2006

Author's response to reviews:

Dr Lutale, JJK
Institute of Medicine University of Bergen
Haraldsplass Hospital,
Ulriksdal 8, N 5009 Bergen
Norway

26-Dec-06

The Editor in Chief
BMC Nephrology.
Revised SUBMISSION OF RESEARCH PAPER TITLED
Microalbuminuria among Type 1 and Type 2 diabetic African patients in Dar es Salaam, Tanzania (MS 8832468231075543)

Authors

J.J.K. Lutale, Janet.Lutale@rasmus.uib.no, jlutale@muchs.ac.tz
H. Thordarson, hraf@haukeland.no
Z. Gulam-Abbas, zabbas@muchs.ac.tz
K. Vetvik, kaare.vetvik@med.uib.no

Version 3: Date 26 December 2006

The Editor

We have responded to changes suggested and the appropriate changes made in the manuscript.

We are now submitting a revised version of the manuscript

sincerely

Lutale JJK
On behalf of the authors